Author's response to reviews

Title: Shamba Maisha: A pilot study assessing Impacts of a micro-irrigation intervention on the health and economic wellbeing of HIV patients

Authors:

Jay A Pandit (jay.pandit@ucsf.edu)
Nicole Sirotin (nicolesirotin@gmail.com)
Robin Tittle (robin.tittle@ucsf.edu)
Elijah Onjolo (elijah.onjolo@kickstart.org)
Elizabeth A Bukusi (ebukusi5@yahoo.com)
Craig R Cohen (ccohen@globalhealth.ucsf.edu)

Version: 4 Date: 5 May 2010

Author's response to reviews: see over
May 03, 2010
BioMed Central Public Health Journal
London, United Kingdom


Dear Editor,

On behalf of my coauthors and myself, I would like to thank you for editorially accepting our manuscript. Below you will find a point-by-point response to the reviewer’s comments and suggested revisions.

- Author list: the list of authors in the manuscript should be written exactly as it is in the submission system, both in style and order. The preferred style is ?First name Initial Last name? (e.g. Joe F Bloggs). Please also remove the authors’ qualifications (e.g. PhD / MD) or job titles from the manuscript and submission system.

Done. (Page 1)

- Authors’ contributions: Please indicate that all authors read and approved the final manuscript.

Done. (Page 14).

- References: All references must be cited in the text, using consecutive numbers in square brackets. Please do not use superscript numbers.

Done.

- Tables: The manuscript cites tables 1, 2, 4 and 5. Table 3 needs to be cited. Furthermore, though tables 4 and 5 are cited, there are no tables 4 and 5 in the manuscript, only 3 are provided. Please amend.

Thank you for picking up on this error. Table 3 has now been cited on Page 8. There are only three tables in the manuscript and therefore I have deleted the citations to table 4 and 5.

- Manuscript: this manuscript does not have the correct structure. Please reformat it to use these headings:

Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors’ contributions; Authors? information (optional); Acknowledgements (optional); References
Done. The discussion section has now been split into a discussion section and a conclusion section.

Minor revisions:

- Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, ?The Title of My Prestigious Publication: A Scientific Study.? should be changed to ?The title of my prestigious publication: a scientific study?. Remember to update the submission system to make sure the title remains consistent.

Done.

Affiliations: Please remember to include the country in all affiliation details

Done.

Title page: Please remove word counts

Done.

Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com

Done.

Abstract: Please remove key words.

Done.

Underlining: Please note that underlining is not supported in the final full text; you may wish to change it to bold.

Done.

References: the reference list should contain up to 30 authors? names for each citation. The term ?et al.? should not be used if there are fewer than 30 authors.

Done.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Done.

Sincerely,

Jay Pandit,
Corresponding Author.